

# **Geophysical Research Letters**<sup>®</sup>

# **RESEARCH LETTER**

10.1029/2021GL094672

### **Key Points:**

- · Forest secondary succession is the main driver of streamflow trends in mountain Mediterranean areas
- The effects of vegetation changes on water availability strongly differ between dry and humid periods
- Trends in streamflow in response to vegetation changes are mostly recorded during the dry and warm season

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

S. M. Vicente-Serrano. svicen@ipe.csic.es

#### Citation:

Castro, F., Murphy, C., Peña-Angulo, D., Tomas-Burguera, M., Noguera, I., et al. (2021). Increased vegetation in mountainous headwaters amplifies water stress during dry periods. Geophysical Research Letters, 48, e2021GL094672. https://doi. org/10.1029/2021GL094672

Received 24 JUN 2021 Accepted 6 SEP 2021

# 1. Introduction

resources occurring during dry periods.

The partitioning of precipitation between blue water, defined as runoff generation, and green water, representing water consumption by vegetation, determines the availability of surface water resources for human activities and freshwater ecosystems (Rulli et al., 2013). Green water is the largest fraction globally (Wang & Dickinson, 2012), but is challenging to quantify (Mueller et al., 2013). Modeling studies suggest a general increase in green water in recent decades, as a consequence of higher plant leaf area (Forzieri et al., 2020; Zeng et al., 2018), longer vegetative periods (Lian et al., 2020), and greater atmospheric evaporative demand (AED) (Vicente-Serrano, McVicar et al., 2020).

**Increased Vegetation in Mountainous Headwaters** 

S. M. Vicente-Serrano<sup>1</sup>, F. Domínguez-Castro<sup>2,3</sup>, C. Murphy<sup>4</sup>, D. Peña-Angulo<sup>1</sup>,

M. Tomas-Burguera<sup>1</sup>, I. Noguera<sup>1</sup>, J. I. López-Moreno<sup>1</sup>, C. Juez<sup>1</sup>, S. Grainger<sup>4</sup>, L. Eklundh<sup>5</sup>,

<sup>2</sup>Aragonese Agency for Research and Development Researcher (ARAID), Zaragoza, Spain, <sup>3</sup>Department of Geography,

<sup>1</sup>Instituto Pirenaico de Ecología, Consejo Superior de Investigaciones Científicas (IPE-CSIC), Zaragoza, Spain,

University of Zaragoza, Zaragoza, Spain, <sup>4</sup>Department of Geography, Irish Climate Analysis and Research UnitS

(ICARUS), Maynooth University, Maynooth, Ireland, <sup>5</sup>Department of Geography, Lund University, Lund, Sweden,

(CIDE-CSIC), Valencia, Spain, 8Department of Geography, Mansoura University, Mansoura, Egypt, 9Department of

<sup>6</sup>Potsdam Institute for Climate Impact Research, Potsdam, Germany, <sup>7</sup>Centro de Investigaciones sobre Desertificación

**Abstract** The dynamics of blue and green water partitioning under vegetation and climate change, as well as their different interactions during wet and dry periods, are poorly understood in the literature. We analyzed the impact of vegetation changes on blue water generation in a central Spanish Pyrenees basin

undergoing intense afforestation. We found that vegetation change is a key driver of large decreases in

to increased plant water consumption. Our findings highlight the importance of vegetation changes in

reinforcing the decrease in water resource availability. With aridity expected to rise in southern Europe

over the next few decades, interactions between climate and land management practices appear to be

Plain Language Summary Forest recovery, due to human land abandonment, has been

to explore how these changes affect resource availability in water-stressed areas. Forest regeneration has resulted in a large decrease in streamflow in a natural catchment in southwestern Europe, which

observed in several regions worldwide. To improve integrated land and water management, it is crucial

cannot be explained by climate change. During the dry season, forest regeneration amplifies the impact

of drought and water availability, with less impacts in the wet season. Therefore, the effects of vegetation recovery on water resources differ based on water availability, with the most serious implications for water

blue water availability. The effect of vegetation increase is amplified during dry years, and mainly during the dry season, with streamflow reductions of more than 50%. This pattern can be attributed primarily

**Amplifies Water Stress During Dry Periods** 

T. Conradt<sup>6</sup>, C. Azorin-Molina<sup>7</sup>, and A. El Kenawy<sup>8,9</sup>

Geography, Sultan Qaboos University, Muscat, Oman

amplifying future hydrological drought risk in the region.

The total vegetation coverage controls the relationship between total evaporation and total precipitation at the catchment scale (Zhang et al., 2001). This would explain how hydrological processes are impacted by changes in leaf area index and plant biomass (Forzieri et al., 2020; Zeng et al., 2018) and the replacement of plant species through secondary succession (Leuschner & Rode, 1999). Studies indicate that reduced tree coverage increases runoff generation after disturbances (Bosch & Hewlett, 1982) since after reduction of the dominant vegetation of a catchment, evaporation is usually reduced (Anderegg et al., 2016; Wine et al., 2017; Winkler et al., 2017). Overall, re-afforestation practices and natural secondary succession reduce runoff production (Filoso et al., 2017), although the magnitude of change is highly dependent on the

© 2021. American Geophysical Union. All Rights Reserved.

Vicente-Serrano, S. M., Domínguez-

vegetation types and the environmental conditions, such as average precipitation (Brown et al., 2005) and forest age (Teuling & Hoek van Dijke, 2020).

In southern Europe, different studies have shown a general reduction of streamflow in recent decades (Gudmundsson et al., 2017; Lorenzo-Lacruz et al., 2012; Stahl et al., 2010). Land abandonment and/or re-afforestation have resulted in a large increase in vegetation coverage in the region's headwaters (Lasanta-Martínez et al., 2005; Lasanta et al., 2017; Sanjuán et al., 2018) and different studies have stressed the fundamental role of higher plant transpiration in explaining streamflow trends in the region (García-Ruiz et al., 2011; Teuling et al., 2019; Vicente-Serrano et al., 2019). Thus, in the Mediterranean mountainous areas of Spain, the increase in the forest surface is the most plausible explanation for streamflow reductions in the headwaters (Beguería et al., 2003; Buendia et al., 2016; Martínez-Fernández et al., 2013; Morán-Tejeda et al., 2012).

Unraveling the interaction between vegetation and climate variability, as well as their impact on the partitioning of precipitation into blue and green water, is a high-priority research topic. While an increase in green water consumption has been linked to greening (Forzieri et al., 2020; Ukkola et al., 2016), it is unclear how water availability and seasonality affect the dependency between vegetation changes and precipitation partitioning. Although some studies have identified a dominance in the green water in response to drought events (Orth & Destouni, 2018), particularly during warm years (Mastrotheodoros et al., 2020), there is little understanding of the interaction between vegetation changes and the interannual and interannual variability of climate conditions to explain anomalies and long-term streamflow trends.

We hypothesize that dominant re-vegetation changes in mountain Mediterranean areas of southern Europe have been the primary cause of the large decline in streamflow observed in recent decades. Nevertheless, the role of these vegetation changes in the partition of precipitation between blue and green water is dependent on interannual climate variability, with the role being stronger during dry years when the system has less available water. Moreover, the role of the vegetation changes in the blue and green partition would be seasonally dependent, with the role of vegetation being stronger in summer dry season, when vegetation is more active.

Here, we analyze the influence of vegetation changes on blue water generation in a humid natural basin (the upper Aragón basin) over the last six decades (1962–2019). This basin, located in the Spanish Pyrenees (2181 km<sup>2</sup>), is characterized by intense secondary succession toward more mature vegetation communities (See Details in Text S1), representing a "typical" example of recent observed vegetation changes in mountainous areas of southern Europe (García-Ruiz et al., 2011; Lasanta et al., 2017). Accordingly, results of this work can be applied to a broad spatial region in Southern Europe, where water scarcity is a serious socioeconomic and environmental issue.

### 2. Material and Methods

#### 2.1. Data

Daily streamflow data for the basin was provided by the Ebro Basin Management Agency (*Confederación Hidrográfica del Ebro* (CHE), http://www.chebro.es/; last accessed February 1, 2021). We derived the streamflow (in cubic hectometers, Hm<sup>3</sup>) generated over the entire basin draining to the Yesa reservoir, which has been calculated by the CHE since 1962 by means of a mass balance using the reservoir data at the daily scale. The accuracy of this mass balance calculation was verified using monthly streamflow data from five available gauging stations spanning the study domain (Figure S3). Data were summarized for the wet (November-June) and dry (July-October) seasons, as well as annually (Figure S4). Herein, hydrological year spans the period between October and September.

Climate data were obtained from a high-resolution (1.1 km) weekly gridded data set for the whole of Spain from 1962 to 2019. This data set was developed using the most complete register of meteorological data provided by the Spanish National Meteorological Service (Agencia Estatal de Meteorología (AEMET), http://www.aemet.es/es/portada; last accessed 1 February 2021). The raw data were quality controlled, homogenized, and interpolated to a common grid resolution of 1.1 km. Further details about this data set development are outlined in Vicente-Serrano, Tomas-Burguera et al. (2017). The gridded data of air temperature, relative humidity, sunshine duration (as a proxy for solar radiation), and wind speed were used to calculate

atmospheric evaporative demand (AED) by means of the FAO-56 Penman-Monteith equation (Pereira et al., 2015). The complete drainage area, which was obtained from a digital elevation model at a spatial resolution of 100 m using ArcHydrotools in ArcGIS 10.2, was used to aggregate monthly precipitation and AED series over the entire basin. To be comparable with streamflow data, climate data were transformed to Hm<sup>3</sup> using the total basin area.

Land cover maps for the decades of 1960 and 2010 were provided by the Spanish Ministry of Agriculture (https://www.mapa.gob.es/es/cartografia-y-sig/publicaciones/agricultura/mac\_2000\_2009.aspx; last accessed February 1, 2021). Based on interpreting aerial photographs and conducting fieldwork, these maps were created at a spatial scale of 1:50000. To illustrate possible changes in greening conditions, the Normal-ized Difference Vegetation Index (NDVI) was calculated at a spatial resolution of 1.1 km using the NOAA-AVHRR images covering the period from 1981 to 2015 (Vicente-Serrano, Martín-Hernández et al., 2020) combined with MODIS NDVI (https://modis.gsfc.nasa.gov/data/dataprod/mod13.php) for the period 2000–2019. Both datasets were standardized using the reference period 2000–2015.

#### 2.2. Methods

We have analyzed the magnitude of the trend in annual P, Q and AED to quantify the magnitude of the annual Q trend that may be associated to climate trends and vegetation changes. Significance of trends in hydrological and climatic variables was analyzed by means of a modified Mann-Kendall trend test, which returns the corrected p values after accounting for temporal pseudoreplication (Hamed & Ramachandra Rao, 1998). To assess the magnitude of change in the different variables, we used a linear regression analysis between the series of time (independent variable) and the climatic and hydrological series (dependent variable). To quantify the magnitude of the annual Q trend that may be associated with climate trends and vegetation changes, we analyzed the annual trend magnitude of P, Q and AED. To figure out how climate and vegetation changes affect annual streamflow, we used multiple regression with streamflow as the dependent variable and precipitation, AED, and time as the independent variables (Beguería et al., 2003). Time was included in the models as a proxy for the progressive evolution of vegetation in the basin as a consequence of secondary succession. Due to the fact that the series only began in 1981, including NDVI was not possible; however, NDVI shows a clear linear increase (Figure S2), indicating that time can be used as a surrogate.

The forward stepwise selection of predictors was used in the construction of regression models using a threshold of 0.05 (Hair et al., 1995). For the annual streamflow analysis, annual precipitation totals and annual AED were considered as independent variables. Seasonally, there are large differences in the time windows in which climate variables are accumulated and affect streamflow, since they depend on physiographic and climatic variables (Barker et al., 2016; López-Moreno et al., 2013). For this reason, we first determined the precipitation and AED cumulative periods with the strongest correlations to streamflow (Figure S5). For the wet and dry seasons, higher correlations were obtained with precipitation accumulated for 10 months before the end of the season. Therefore, the wet season precipitation totals from September to June were used as the independent variable in the regression model, while the dry season precipitation totals from January to October were used as the dependent variable. For the wet season, the maximum correlation between AED and streamflow was obtained for nine months, and for the dry season, the maximum correlation was obtained for 10 months.

The ordinary least square regression method was used to assess trends for wet and dry years and seasons. Herein, wet years (seasons) were defined as those exceeding the 70th percentile of precipitation over the period of record. On the other hand, dry years (seasons) were determined as those with rainfall falling below the 30th percentile.

#### 3. Results and Discussion

Annual precipitation in the basin shows high variability, with a non-significant decrease of 8% in annual totals observed between 1962 and 2019 (Figure 1). Over the same time period, AED increased by 5.7%, which is statistically significant at the 95% level (p < 0.05). From 1962 to 2019, the annual streamflow in the catchment decreased by more than 40%, which is consistent with the pattern observed in other natural non-managed basins in Spain (Martínez-Fernández et al., 2013). The decrease in annual streamflow is close





**Figure 1.** Temporal evolution of annual precipitation (a), atmospheric evaporative demand (AED) (b) and streamflow (c) in the upper Aragón basin for 1962–2019. Dashed lines represent the linear trend obtained by means of least-squares. The percentage changes between 1961 and 2019 are obtained from the regression lines. P-values are obtained by means of the modified Mann-Kendall test (See Section 2.2).

to the increase (34.1%) in the estimated evaporation by means of the annual water balance. The decrease in streamflow cannot be attributed to changes in precipitation. Our findings indicate that annual precipitation decreased by 239.6 Hm<sup>3</sup> over the 58-year study period, while annual streamflow declined by 692.5 Hm<sup>3</sup>. This difference (452.85 Hm<sup>3</sup>) can be attributed to the net increase in evaporation. An assumption can be made that the amount of increase in AED between 1962 and 2019 corresponds to Evapotranspiration (E), as consequence of direct radiative forcing associated to warming. Nonetheless, this increase is only 130.9 Hm<sup>3</sup> and there are still 321 Hm<sup>3</sup> of streamflow decline between 1962 and 2019 that would not be explained by the observed climate trend. Moreover, it is important to note that the increase in observed AED cannot be completely associated to the increase in E since during summer months soil moisture deficits are common and there is an evaporation deficit (E-AED), which is the main driver controlling forest growth in the region (Vicente-Serrano et al., 2015). Thus drought has significant consequences on vegetation activity and growth in the study area (Camarero et al., 2011; Pasho et al., 2011; Peguero-Pina et al., 2007; Vicente-Serrano et al., 2021). After removing the role of precipitation, the non-climate-related streamflow decline would range between 321 Hm<sup>3</sup> (if AED increase is fully representing an increase in E) and 452.8 Hm<sup>3</sup> (assuming E changes are not driven by AED). These numbers represent between 46% and 65% of total streamflow decline, which would be explained primarily by the basin's secondary succession process.

The secondary role of the observed increase in AED in the streamflow reduction is confirmed by the regression analysis. Over the period 1962–2019, regression analysis using precipitation, AED, and time (in years, as a proxy for the effect of secondary succession and general greenness) as predictors of annual streamflow

#### Table 1

Correlation Coefficients of the Linear Stepwise Regression Analysis in Which Annual and Seasonal Streamflow was the Dependent Variable and Precipitation, Atmospheric Evaporative Demand and Time (in Years) Were the Independent Predictors

		Partial correlations		
	$\mathbb{R}^2$	Precipitation	AED	Time
Annual	0.82	0.88	-0.09 (n.s.)	-0.64
Wet season (November–June)	0.87	0.92	-0.06 (n.s.)	-0.63
Dry season (July–October)	0.48	0.31	-0.21	-0.49

*Note.*  $R^2$  represents the percentage of total variance of the independent variable explained by the predictors. Partial correlations represent the role of each independent variable, removing the influence of the covariates. Predictors showing a non-significant (0.05 level) role in explaining streamflow variability were not included in the final models.

explains 82% of the variability in streamflow. However, AED was removed from our final model because it was not a significant predictor (Table 1). According to the partial correlation, precipitation has the largest (positive) influence on annual streamflow, while time exhibited a significant (negative) correlation. Annual precipitation showed a significant negative correlation with annual AED (Figure S6). The role of cloudiness in reducing solar radiation and air temperature can explain this negative relationship between precipitation and AED. Notably, the correlation between annual AED and streamflow was statistically non-significant after fixing (controlling) precipitation effect.

Seasonally, there is also a large decrease in streamflow, being much stronger (-63.7%) during the dry season (Figure S7). Again, this decrease can not be driven by precipitation, which showed a statistically non-significant decrease on the seasonal scale. During the dry season, AED showed a significant negative partial correlation with streamflow, but its influence on streamflow is minor compared to precipitation and time. In the dry season, the inclusion of AED in the linear model accounted only for 2.3% of the total variance in streamflow.

It can be concluded that while the interannual variability of streamflow is strongly correlated with annual precipitation totals, the decrease in streamflow cannot be explained by either precipitation decrease or enhanced AED. Rather, decreasing streamflow trends can be mostly associated with the progressive increase in vegetation cover and greenness over time. Previous studies suggested that the effect of increased greenness on precipitation partitioning between blue and green water is more pronounced in dry areas (Ukkola et al., 2016; Zeng et al., 2018). Based on empirical observations from the upper Aragón basin, where average precipitation is 1,300 mm year<sup>-1</sup>, our findings suggest that in a humid region affected by secondary succession and increased greenness, vegetation changes can also play a key role in reducing blue water.

Furthermore, we found the effect of secondary succession on the partitioning of precipitation to be strongly differential between wet and dry years. Both precipitation and AED show similar non-significant trends during wet and dry years (Figure S8). However, a small (-15.4%) and non-significant decrease in annual streamflow was found in wet years, compared to a larger (-52.1%) and statistically significant decrease during dry years (Figure 2). The role of AED on streamflow evolution is non-statistically significant during either wet or dry years (Table S1). In wet years, green water increase is expected as a consequence of higher water consumption by more dense vegetation coverage but during these years, the abundance of precipitation makes that secondary succession and greenness have less impact on streamflow trend. In contrast, in dry years secondary succession tends to reduce blue water, in favor of increased green water use. Thus, blue water has been shown to decrease more during dry than wet years in Europe (Orth & Destouni, 2018), so it would be expected that the effect of secondary succession and increased greenness would be amplified during dry years. Vegetation tends to adapt maximum transpiration to available soil moisture (Grossiord et al., 2020), consuming water necessary for physiological processes as the first "ecosystems priority" by increasing water use efficiency (Peters et al., 2018) and, consequently, reducing blue water generation. The differential effect of greenness on streamflow reduction between dry and humid regions (Ukkola et al., 2016; Zeng et al., 2018) identified by previous studies, is shown in the upper Aragón basin between wet and dry





Figure 2. Evolution of annual streamflow during wet (a) and dry (b) precipitation years.

years; in wetter years the greater availability of water in the system, generates proportionally more blue water than during dry years.

The influence of secondary succession also differs between wet and dry seasons (Figure 3). Precipitation and AED reveal non-significant changes in the wet season (November-June) in both wet and dry seasons (Figure S9). Similarly, no significant decrease in streamflow is found since the decrease of blue water during dry seasons is not statistically significant. In the upper Aragón basin vegetation activity shows strong seasonality. As a consequence of cold air temperatures and dominant snow coverage in winter, vegetation activity is low during these months. Therefore, although secondary succession contributes to some decrease in blue water generation in the wet season, particularly during dry periods (Table S1), the main effect is recorded during the warmer dry season (July-October), which is characterized by high vegetation activity, and consequently water consumption over the different altitudinal belts that characterize vegetation in the basin (García-Ruiz et al., 2015). A significant decrease in streamflow during the dry season is evident in both dry and wet periods, with declines being more pronounced in rainy periods. Even in wet periods, the decrease in blue water during the dry season has been significant, probably as consequence of increased plant coverage and greenness. Consequently the decrease in dry season blue water generation has been dramatic during both dry and wet periods. Increased plant water needs as a consequence of enhanced AED and competition for available soil moisture not only cause strong decreases in blue water but also episodes of plant stress, low plant growth and forest dieback in response to drought (Vicente-Serrano et al., 2021), which have become reinforced due to vegetation changes, climate variability and change in recent decades (Camarero et al., 2011; Macias et al., 2006).

Future climate change scenarios over the Mediterranean region show decreases in precipitation and more frequent droughts (Lionello & Scarascia, 2018). Large increases in AED are also projected (Vicente-Serrano, McVicar et al., 2020). Under these scenarios, it is expected that blue water generation will decrease further in large areas of southern Europe affected by land abandonment (Lasanta et al., 2017), plant colonization and secondary succession (García-Ruiz et al., 2011). However, large uncertainties are associated with trends in future vegetation cover. While vegetation changes in the region have been considerable to date, potential remains for significant future changes, including the replacement of coniferous forests and shrublands by broadleaf forests below elevations of 1,600 m a.s.l., and the advance of shrublands and forests at elevations above 2,000 m a.s.l., as consequence of the pasture abandonment and increases in air temperature (Sanjuán et al., 2018). This evolution could affect future availability of water resources since once the current dominant young forests reach more mature levels they may demand more water, as old forests have a lower water yield response (Teuling & Hoek van Dijke, 2020). Although increased atmospheric  $CO_2$  concentrations could have limited plant water needs (Swann, 2018), our results suggest that water availability for so-cio-economic activities have been primarily driven by water demands from vegetation changes. This finding concur with modeling studies at larger scales (Mankin et al., 2019).





# Humid season (November-June)



# Dry season (July-October)



Figure 3. Evolution of annual streamflow during the wet (a) and dry (b) seasons during high and low precipitation periods.

Mountainous areas in southern Europe are the water towers (Viviroli & Weingartner, 2004) in which the usable water resources of the region are generated. Water supply to irrigated areas is already restricted during dry years in the study domain (Vicente-Serrano, Zabalza-Martínez et al., 2017, Vicente-Serrano et al., 2021). These findings, coupled with projected land use and climate change in southern Europe suggest that sustainable land management practices must be focused on limiting green water losses to enhance availability of blue water into the future. In this light, practices such as thinning (Manrique-Alba et al., 2020) and clearing shrublands for livestock (Lasanta et al., 2019) could be considered sustainable water management practices in southern Europe (García-Ruiz et al., 2020).

## 4. Conclusions

This study analyzed the partition of the available water resources between blue and green water in a Spanish mountainous Mediterranean basin from 1960 to 2019. The study basin is characterized by substantial plant secondary succession processes. Overall, the main conclusions of this study can be summarized: (a) blue water has declined by 40% in the study domain over the last six decades, (b) climate trends accounted for a small portion of blue water reduction, (c) the strong decrease in streamflow can be explained largely by the process of plant secondary succession and increased vegetation, (d) the partition of precipitation trends



between blue and green water trends differs between dry and wet years, (e) there is stronger increase in the total basin evaporation during dry years, which drastically limits the production of blue water, posing major challenges to water availability during droughts episodes and (f) there are significant seasonal differences in the role of dry and humid years in precipitation partitioning between blue and green water. The most pronounced impacts can be noted during dry season, when vegetation is more active and consume more water, reducing dramatically water resources, even in wet summers.

#### **Data Availability Statement**

The data used in this study can be downloaded at: https://doi.org/10.5281/zenodo.5287430.

#### References

- Anderegg, W. R. L., Martinez-Vilalta, J., Cailleret, M., Camarero, J. J., Ewers, B. E., Galbraith, D., et al. (2016). When a tree dies in the forest: Scaling climate-driven tree mortality to ecosystem water and carbon fluxes. *Ecosystems*, 19(6), 1133–1147. https://doi.org/10.1007/ s10021-016-9982-1
- Barker, L. J., Hannaford, J., Chiverton, A., & Svensson, C. (2016). From meteorological to hydrological drought using standardised indicators. *Hydrology and Earth System Sciences*, 20(6), 2483–2505. https://doi.org/10.5194/hess-20-2483-2016
- Beguería, S., López-Moreno, J. I., Lorente, A., Seeger, M., & García-Ruiz, J. M. (2003). Assessing the effect of climate oscillations and land-use changes on streamflow in the Central Spanish Pyrenees. *Ambio*, *32*(4), 283–286. https://doi.org/10.1579/0044-7447-32.4.283
- Bosch, J. M., & Hewlett, J. D. (1982). A review of catchment experiments to determine the effect of vegetation changes on water yield and evapotranspiration. *Journal of Hydrology*, 55(1), 3–23. https://doi.org/10.1016/0022-1694(82)90117-2
- Brown, A. E., Zhang, L., McMahon, T. A., Western, A. W., & Vertessy, R. A. (2005). A review of paired catchment studies for determining changes in water yield resulting from alterations in vegetation. *Journal of Hydrology*, 310(1), 28–61. https://doi.org/10.1016/j. jhydrol.2004.12.010
- Buendia, C., Bussi, G., Tuset, J., Vericat, D., Sabater, S., Palau, A., & Batalla, R. J. (2016). Effects of afforestation on runoff and sediment load in an upland Mediterranean catchment. *Science of the Total Environment*, 540, 144–157. https://doi.org/10.1016/j.scitotenv.2015.07.005 Camarero, J. J., Bigler, C., Linares, J. C., & Gil-Pelegrín, E. (2011). Synergistic effects of past historical logging and drought on the decline of
- Pyrenean silver fir forests. Forest Ecology and Management, 262(5), 759-769. https://doi.org/10.1016/j.foreco.2011.05.009
- Filoso, S., Bezerra, M. O., Weiss, K. C. B., & Palmer, M. A. (2017). Impacts of forest restoration on water yield: A systematic review. *PloS One*, *12*, e0183210. https://doi.org/10.1371/journal.pone.0183210
- Forzieri, G., Miralles, D. G., Ciais, P., Alkama, R., Ryu, Y., Duveiller, G., et al. (2020). Increased control of vegetation on global terrestrial energy fluxes. *Nature Climate Change*, 10(4), 356–362. https://doi.org/10.1038/s41558-020-0717-0
- García-Ruiz, J. M., Lasanta, T., Nadal-Romero, E., Lana-Renault, N., & Álvarez-Farizo, B. (2020). Rewilding and restoring cultural landscapes in Mediterranean mountains: Opportunities and challenges. Land Use Policy. (Vol. 99, p. 104850). https://doi.org/10.1016/j. landusepol.2020.104850
- García-Ruiz, J. M., López-Moreno, J. I., Lasanta, T., Vicente-Serrano, S. M., González-Sampériz, P., Valero-Garcés, B. L., et al. (2015). Geo-ecological effects of global change in the Central Spanish Pyrenees: A review at different spatial and temporal scales [Los efectos geoecológicos del cambio global en el pirineo central español: Una revisión a distintas escalas espaciales y temporales. *Pirineos, 170*, e012. https://doi.org/10.3989/Pirineos.2015.170005
- García-Ruiz, J. M., López-Moreno, J. I., Vicente-Serrano, S. M., Lasanta-Martínez, T., & Beguería, S. (2011). Mediterranean water resources in a global change scenario. *Earth-Science Reviews*, 105(3–4), 121–139. https://doi.org/10.1016/j.earscirev.2011.01.006
- Grossiord, C., Buckley, T. N., Cernusak, L. A., Novick, K. A., Poulter, B., Siegwolf, R. T. W., et al. (2020). Plant responses to rising vapor pressure deficit. New Phytologist, 226(6), 1550–1566. https://doi.org/10.1111/nph.16485
- Gudmundsson, L., Seneviratne, S. I., & Zhang, X. (2017). Anthropogenic climate change detected in European renewable freshwater resources. Nature Climate Change, 7(11), 813–816. https://doi.org/10.1038/nclimate3416
- Hair, J. F., Anderson, R. E., Tatham, R. L., & Black, W. C. (1995). No title. Multivariate data analysis.
- Hamed, K. H., & Ramachandra Rao, A. (1998). A modified Mann-Kendall trend test for autocorrelated data. *Journal of Hydrology*, 204(1–4), 182–196. https://doi.org/10.1016/S0022-1694(97)00125-X
- Lasanta, T., Arnáez, J., Pascual, N., Ruiz-Flaño, P., Errea, M. P., & Lana-Renault, N. (2017). Space-time process and drivers of land abandonment in Europe. Catena, 149, 810–823. https://doi.org/10.1016/j.catena.2016.02.024
- Lasanta, T., Nadal-Romero, E., & García-Ruiz, J. M. (2019). Clearing shrubland as a strategy to encourage extensive livestock farming in the Mediterranean Mountains. *Geographical Research Letters*, 45(2), 487–513. https://doi.org/10.18172/cig.3616
- Lasanta-Martínez, T., Vicente-Serrano, S. M., & Cuadrat-Prats, J. M. (2005). Mountain Mediterranean landscape evolution caused by the abandonment of traditional primary activities: A study of the Spanish Central Pyrenees. *Applied Geography*, 25(1), 47–65. https://doi. org/10.1016/j.apgeog.2004.11.001
- Leuschner, C., & Rode, M. W. (1999). The role of plant resources in forest succession: Changes in radiation, water and nutrient fluxes, and plant productivity over a 300-yr-long chronosequence in NW-Germany. *Perspectives in Plant Ecology, Evolution and Systematics*, 2, 103–147. https://doi.org/10.1078/1433-8319-00067
- Lian, X., Piao, S., Li, L. Z. X., Li, Y., Huntingford, C., Ciais, P., et al. (2020). Summer soil drying exacerbated by earlier spring greening of northern vegetation. *Science Advances*, 6(1), eaax0255. https://doi.org/10.1126/sciadv.aax0255
- Lionello, P., & Scarascia, L. (2018). The relation between climate change in the Mediterranean region and global warming. Regional Environmental Change, 18(5), 1481–1493. https://doi.org/10.1007/s10113-018-1290-1
- López-Moreno, J. I., Vicente-Serrano, S. M., Zabalza, J., Beguería, S., Lorenzo-Lacruz, J., Azorin-Molina, C., & Morán-Tejeda, E. (2013). Hydrological response to climate variability at different time scales: A study in the Ebro basin. *Journal of Hydrology*, 477, 175–188. https://doi.org/10.1016/j.jhydrol.2012.11.028
- Lorenzo-Lacruz, J., Vicente-Serrano, S. M., López-Moreno, J. I., Morán-Tejeda, E., & Zabalza, J. (2012). Recent trends in Iberian streamflows (1945-2005). Journal of Hydrology, 414–415. https://doi.org/10.1016/j.jhydrol.2011.11.023

#### Acknowledgments

This work was supported by projects CGL2017-82216-R, PCI2019-103631, and PID2019-108589RA-I00 financed by the Spanish Commission of Science and Technology and FEDER; CROSSDRO project financed by AXIS (Assessment of Cross(X)-sectoral climate Impacts and pathways for Sustainable transformation), JPI-Climate co-funded call of the European Commission and INDECIS which is part of ERA4CS, an ERA-NET initiated by JPI Climate, and funded by FORMAS (SE), DLR (DE), BMWFW (AT), IFD (DK), MINECO (ES), ANR (FR) with co-funding by the European Union (Grant 690462). Dhais Peña-Angulo received a "Juan de la Cierva" postdoctoral contract (FJCI-2017-33652 Spanish Ministry of Economy and Competitiveness, MEC). Miquel Tomas-Burguera received a "Juan de la Cierva" postdoctoral contract (FJCI-2019-039261-I Spanish Ministry of Science and Innovation). C. Azorin-Molina and S. Grainger. acknowledge funding from the Irish Environmental Protection Agency grant 2019-CCRP-MS.60. C. Juez acknowledges funding from the H2020-MSCA-IF-2018 programme (Marie Sklodowska-Curie Actions) of the European Union under REA grant agreement, number 834329-SEDILAND.

- Macias, M., Andreu, L., Bosch, O., Camarero, J. J., & Gutiérrez, E. (2006). Increasing aridity is enhancing silver fir (Abies alba Mill.) water stress in its south-western distribution limit. *Climatic Change*, 79(3–4), 289–313. https://doi.org/10.1007/s10584-006-9071-0
- Mankin, J. S., Seager, R., Smerdon, J. E., Cook, B. I., & Williams, A. P. (2019). Mid-latitude freshwater availability reduced by projected vegetation responses to climate change. *Nature Geoscience*, *12*(12), 983–988. https://doi.org/10.1038/s41561-019-0480-x
- Manrique-Alba, À., Beguería, S., Molina, A. J., González-Sanchis, M., Tomàs-Burguera, M., del Campo, A. D., et al. (2020). Long-term thinning effects on tree growth, drought response and water use efficiency at two Aleppo pine plantations in Spain. Science of the Total Environment, 728, 138536. https://doi.org/10.1016/j.scitotenv.2020.138536
- Martínez-Fernández, J., Sánchez, N., & Herrero-Jiménez, C. M. (2013). Recent trends in rivers with near-natural flow regime: The case of the river headwaters in Spain. *Progress in Physical Geography*, *37*(5), 685–700. https://doi.org/10.1177/0309133313496834
- Mastrotheodoros, T., Pappas, C., Molnar, P., Burlando, P., Manoli, G., Parajka, J., et al. (2020). More green and less blue water in the Alps during warmer summers. *Nature Climate Change*, 10(2), 155–161. https://doi.org/10.1038/s41558-019-0676-5
- Morán-Tejeda, E., Ceballos-Barbancho, A., Llorente-Pinto, J. M., & López-Moreno, J. I. (2012). Land-cover changes and recent hydrological evolution in the Duero Basin (Spain). Regional Environmental Change, 12(1), 17–33. https://doi.org/10.1007/s10113-011-0236-7
- Mueller, B., Hirschi, M., Jimenez, C., Ciais, P., Dirmeyer, P. A., Dolman, A. J., et al. (2013). Benchmark products for land evapotranspiration: LandFlux-EVAL multi-data set synthesis. *Hydrology and Earth System Sciences*, 17, 3707–3720. https://doi.org/10.5194/ hess-17-3707-2013
- Orth, R., & Destouni, G. (2018). Drought reduces blue-water fluxes more strongly than green-water fluxes in Europe. *Nature Communications*, 9(1). https://doi.org/10.1038/s41467-018-06013-7
- Pasho, E., Camarero, J. J., de Luis, M., & Vicente-Serrano, S. M. (2011). Impacts of drought at different time scales on forest growth across a wide climatic gradient in north-eastern Spain. Agricultural and Forest Meteorology, 151(12), 1800–1811. https://doi.org/10.1016/j. agrformet.2011.07.018
- Peguero-Pina, J. J., Camarero, J. J., Abadía, A., Martín, E., González-Cascón, R., Morales, F., & Gil-Pelegrín, E. (2007). Physiological performance of silver-fir (Abies alba Mill.) populations under contrasting climates near the south-western distribution limit of the species. *Flora: Morphology, Distribution, Functional Ecology of Plants, 202*(3), 226–236. https://doi.org/10.1016/j.flora.2006.06.004
- Pereira, L. S., Allen, R. G., Smith, M., & Raes, D. (2015). Crop evapotranspiration estimation with FAO56: Past and future. Agricultural Water Management, 147, 4–20. https://doi.org/10.1016/j.agwat.2014.07.031
- Peters, W., van der Velde, I. R., van Schaik, E., Miller, J. B., Ciais, P., Duarte, H. F., et al. (2018). Increased water-use efficiency and reduced CO<sub>2</sub> uptake by plants during droughts at a continental scale. *Nature Geoscience*, *11*(10), 744–748. https://doi.org/10.1038/ s41561-018-0212-7
- Rulli, M. C., Saviori, A., & D'Odorico, P. (2013). Global land and water grabbing. Proceedings of the National Academy of Sciences, 110, 892–897. https://doi.org/10.1073/pnas.1213163110
- Sanjuán, Y., Arnáez, J., Beguería, S., Lana-Renault, N., Lasanta, T., Gómez-Villar, A., et al. (2018). Woody plant encroachment following grazing abandonment in the subalpine belt: A case study in northern Spain. *Regional Environmental Change*, 18(4), 1103–1115. https:// doi.org/10.1007/s10113-017-1245-y
- Stahl, K., Hisdal, H., Hannaford, J., Tallaksen, L. M., Van Lanen, H. A. J., Sauquet, E., et al. (2010). Streamflow trends in Europe: Evidence from a dataset of near-natural catchments. *Hydrology and Earth System Sciences*, 14(12), 2367–2382. https://doi.org/10.5194/ hess-14-2367-2010
- Swann, A. L. S. (2018). Plants and drought in a changing climate. Current Climate Change Reports, 4(2), 192–201. https://doi.org/10.1007/ s40641-018-0097-y
- Teuling, A. J., de Badts, E., Jansen, F. A., Fuchs, R., Buitink, J., van Dijke, A. J., & Sterling, S. (2019). Climate change, re-/afforestation, and urbanisation impacts on evapotranspiration and streamflow in Europe, 23. Hydrology and Earth System Sciences Discussions, 1–30. https://doi.org/10.5194/hess-2018-634
- Teuling, A. J., & Hoek van Dijke, A. J. (2020). Forest age and water yield. Nature, 578(7794), E16-E18. https://doi.org/10.1038/ s41586-020-1941-5
- Ukkola, A. M., Prentice, I. C., Keenan, T. F., Van Dijk, A. I. J. M., Viney, N. R., Myneni, R. B., & Bi, J. (2016). Reduced streamflow in water-stressed climates consistent with CO<sub>2</sub> effects on vegetation. *Nature Climate Change*, 6(1), 75–78. https://doi.org/10.1038/ nclimate2831
- Vicente-Serrano, S. M., Camarero, J. J., Zabalza, J., Sangüesa-Barreda, G., López-Moreno, J. I., & Tague, C. L. (2015). Evapotranspiration deficit controls net primary production and growth of silver fir: Implications for Circum-Mediterranean forests under forecasted warmer and drier conditions. Agricultural and Forest Meteorology, 206, 45–54. https://doi.org/10.1016/j.agrformet.2015.02.017
- Vicente-Serrano, S. M., Martín-Hernández, N., Reig, F., Azorin-Molina, C., Zabalza, J., Beguería, S., et al. (2020). Vegetation greening in Spain detected from long term data (1981–2015). *International Journal of Remote Sensing*, 41(5), 1709–1740. https://doi.org/10.1080/0 1431161.2019.1674460
- Vicente-Serrano, S. M., McVicar, T. R., Miralles, D. G., Yang, Y., & Tomas-Burguera, M. (2020). Unraveling the influence of atmospheric evaporative demand on drought and its response to climate change. WIREs Climate Change, 11(2), e632. https://doi.org/10.1002/ wcc.632
- Vicente-Serrano, S. M., Peña-Angulo, D., Murphy, C., López-Moreno, J. I., Tomas-Burguera, M., & Domínguez-Castro, F. (2021). The complex multi-sectoral impacts of drought: Evidence from a mountainous basin in the Central Spanish Pyrenees. *Science of the Total Environment*, 769, 144702. https://doi.org/10.1016/j.scitotenv.2020.144702
- Vicente-Serrano, S. M., Peña-Gallardo, M., Hannaford, J., Murphy, C., Lorenzo-Lacruz, J., Dominguez-Castro, F., et al. (2019). Climate, irrigation, and land-cover change explain streamflow trends in countries bordering the Northeast Atlantic. *Geophysical Research Letters*, 46(19), 10821–10833. https://doi.org/10.1029/2019GL084084
- Vicente-Serrano, S. M., Tomas-Burguera, M., Beguería, S., Reig, F., Latorre, B., Peña-Gallardo, M., et al. (2017). A high resolution dataset of drought indices for Spain. *Data*, 2(3). https://doi.org/10.3390/data2030022
- Vicente-Serrano, S. M., Zabalza-Martínez, J., Borràs, G., López-Moreno, J. I., Pla, E., Pascual, D., et al. (2017). Extreme hydrological events and the influence of reservoirs in a highly regulated river basin of northeastern Spain. *Journal of Hydrology: Regional Studies*, 12, 13–32. https://doi.org/10.1016/j.ejrh.2017.01.004
- Viviroli, D., & Weingartner, R. (2004). The hydrological significance of mountains: From regional to global scale. Hydrology and Earth System Sciences, 8(6), 1016–1029. https://doi.org/10.5194/hess-8-1017-2004
- Wang, K., & Dickinson, R. E. (2012). A review of global terrestrial evapotranspiration: Observation, modeling, climatology, and climatic variability. *Reviews of Geophysics*, 50(2), RG2005. https://doi.org/10.1029/2011RG000373

- Wine, M., Cadol, D., & Makhnin, O. (2017). In ecoregions across western USA streamflow increases during post-wildfire recovery. Environmental Research Letters, 13. https://doi.org/10.1088/1748-9326/aa9c5a
- Winkler, R., Spittlehouse, D., & Boon, S. (2017). Streamflow response to clear-cut logging on British Columbia's Okanagan Plateau. *Ecohy*drology, 10(2), e1836. https://doi.org/10.1002/eco.1836
- Zeng, Z., Piao, S., Li, L. Z. X., Wang, T., Ciais, P., Lian, X., et al. (2018). Impact of Earth greening on the terrestrial water cycle. *Journal of Climate*, *31*(7), 2633–2650. https://doi.org/10.1175/JCLI-D-17-0236.1
- Zhang, L., Dawes, W. R., & Walker, G. R. (2001). Response of mean annual evapotranspiration to vegetation changes at catchment scale. *Water Resources Research*, *37*(3), 701–708. https://doi.org/10.1029/2000WR900325